\chapter{使用者問卷調查項目}
\label{ch:survey-items}

項目 \ref{item:pssuq:start} 至 \ref{item:pssuq:end} 來自於 PSSUQ 第三版。由於 PSSUQ 尚未有廣被採納的繁體中文翻譯，在問卷上保留以原文呈現避免翻譯可能造成歧義產生。項目 \ref{item:pssuq:start} 至 \ref{item:sysuse:end} 屬於 SysUse 類別，\ref{item:infoqual:start} 至 \ref{item:infoqual:end} 屬於 InfoQual，\ref{item:intqual:start} 至 \ref{item:intqual:end} 屬於 IntQual。自 \ref{item:custom:start} 起為本研究針對平台另外設計的項目，採用中文方便表達與理解。

\begin{singlespacing}
\begin{enumerate}
    \item \label{item:pssuq:start} Overall, I am satisfied with how easy it is to use this system.
    \item It was simple to use this system.
    \item I was able to complete the tasks and scenarios quickly using this system.
    \item I felt comfortable using this system.
    \item It was easy to learn to use this system.
    \item \label{item:sysuse:end} I believe I could become productive quickly using this system.
    \item \label{item:infoqual:start} The system gave error messages that clearly told me how to fix problems.
    \item Whenever I made a mistake using the system, I could recover easily and quickly.
    \item The information (such as on-line help, on-screen messages and other documentation) provided with this system was clear.
    \item It was easy to find the information I needed.
    \item The information was effective in helping me complete the tasks and scenarios.
    \item \label{item:infoqual:end} The organization of information on the system screens was clear.
    \item \label{item:intqual:start} The interface of this system was pleasant.
    \item I liked using the interface of this system.
    \item \label{item:intqual:end} This system has all the functions and capabilities I expect it to have.
    \item \label{item:pssuq:end} Overall, I am satisfied with this system.
    \item \label{item:custom:start} 平台的容器資源限制預設值足以運行一般工作負載
    \item kubectl 頁面的效能足夠使用（指令互動，不包含環境啟動時間）
    \item kubectl 頁面所佔用的資源配額合理，大致不影響使用者權益
    \item Quota 頁面有助於分析平台配額使用狀況
    \item 比起其他 Kubernetes 客戶端，我會優先使用平台網頁界面
    \item 比起自行安裝 kubectl，我會優先使用網頁界面 kubectl 頁面功能
    \item 比起其他的 Kubernetes 代管服務，我傾向使用正式上線的 CSKloud
    \item 我熟悉 Kubernetes
\end{enumerate}
\end{singlespacing}
